Author's response to reviews

Title: Sigmoid to Scrotal Fistula Secondary to Mesh Erosion: A Rare Complication of Inguinal Hernia Repair in a Patient on Anticoagulation

Authors:

Jad A Degheili (jdegheili@gmail.com)
Maen Aboul-Hosn (jdegheili@gmail.com)
Mustapha El Lakis (jdegheili@gmail.com)
Firas Abiad (jdegheili@gmail.com)
Ali H Hallal (jdegheili@gmail.com)

Version: 2
Date: 20 February 2015

Author's response to reviews:

To,
The Editors,
BMC Surgery-Case Reports

Subject: Resubmission of case report titled: “Sigmoid to Scrotal Fistula Secondary to Mesh Erosion: A Rare Complication of Inguinal Hernia Repair in a Patient on Anticoagulation”

Dear Sir/ Ma’am

Further to your reply, several changes have been made to our manuscript.

These changes include the following:
1-Detailed authors’ information including emails as well.
2-Line numbering of the main text of our manuscript.
3-An "Acknowledgement" and a "Consent" paragraph have been added.
4-Modifying the reference format as per your recommended style, present on the BMC website.
5-A "CASE checklist" has been uploaded
6- The final manuscript version has been re-uploaded as well.

Hope these changes will be up to your expectations.

Kind regards and wishes
Sincerely yours,
Jad Degheili